Author’s response to reviews

Title: Influence of Social Support on Cognitive Function in the Elderly

Authors:

Dr Shu-Chuan Jennifer Yeh (syeh@cm.nsysu.edu.tw)
Yea-Ying Liu (veyili@cc.kmu.edu.tw)

Version: 6 Date: 25 May 2003

Dear Dr. Enitan Sawyerr:

Thank you very much for your e-mail message dated May 21. Now my manuscript has been revised completely and submitted to the center. We added limitation of the validation of social support and loneliness to the text based on Dr. Rait's comments. In addition, all the figure numbers have been checked to ascertain the consistency between text and tables. We did not change the decimal point because one decimal point could not satisfy all the necessary reports. Thank you very much for your consideration.

Sincerely,

Shu-Chuan Jennifer Yeh